Author’s response to reviews

Title: A very early diagnosis of Alström syndrome by next generation sequencing

Authors:

Leonardo Gatticchi (leonardo.gatticchi@gmail.com)
Jan Miertus (director@genius.care)
Paolo Enrico Maltese (paolo.maltese@assomagi.org)
Simone Bressan (simone.bressan@assomagi.org)
Luca De Antoni (luca.deantoni@assomagi.org)
Ludmila Podracká (Ludmila.Podracka@nudch.eu)
Vanda Rísová (vanda.risova@gmail.com)
Mari Mällo (mari.mallo@asperbio.com)
Kaie Jaakson (kaie.jaakson@asperbio.com)
Kairit Joost (kairit.joost@asperbio.com)
Leonardo Colombo (leonardo.colombo.82@gmail.com)
Matteo Bertelli (matteo.bertelli@assomagi.org)

Version: 2 Date: 04 May 2020

Author’s response to reviews:

Dear Publisher,

We added the following sentence to "Consent for participation”:

Written informed consent was obtained from the individual(s), and minor(s)’ legal guardian/next of kin, for the publication of any potentially identifiable images or data included in this article. The misunderstanding arose from the fact that we have only one informed consent form that includes consent to participate and consent to the publication of clinical data. Furthermore, since the paper do not include images of our patients, we thought that under the heading "Consent to publication" one could write "not applicable". We apologize for this misunderstanding.
Sincerely,

Paolo Maltese